# THE ROYAL SOCIETY

# ROYAL SOCIETY OPEN SCIENCE

# The effect of face masks and sunglasses on identity and expression recognition with super-recognizers and typical observers

Eilidh Noyes, Josh P. Davis, Nikolay Petrov, Katie L. H. Gray and Kay L. Ritchie

#### Article citation details

R. Soc. open sci. 8: 201169. http://dx.doi.org/10.1098/rsos.201169

#### Review timeline

Original submission: 7 July 2020 Revised submission: 27 November 2020 Revised submission: 8 February 2021

Final acceptance: 4 March 2021

Note: Reports are unedited and appear as submitted by the referee. The review history

appears in chronological order.

# **Review History**

RSOS-201169.R0 (Original submission)

Review form: Reviewer 1

Is the manuscript scientifically sound in its present form?

Are the interpretations and conclusions justified by the results?

Is the language acceptable?

Yes

Do you have any ethical concerns with this paper?

No

Have you any concerns about statistical analyses in this paper?

No

Reports © 2021 The Reviewers; Decision Letters © 2021 The Reviewers and Editors; Responses © 2021 The Reviewers, Editors and Authors. Published by the Royal Society under the terms of the Creative Commons Attribution License http://creativecommons.org/licenses/by/4.0/, which permits unrestricted use, provided the original author and source are credited

#### Recommendation?

Major revision is needed (please make suggestions in comments)

#### Comments to the Author(s)

Noyes et al. examined the extent to which recognition of famous face identity (E1), unfamiliar face identity (E2), and face emotion (E3) is influenced by disguising the eyes with sunglasses or the mouth with masks. This was compared in people with typical recognition ability and so-called super recognizers (SR). Recognition performance was measured in a face matching task (E1-2) and a forced-choice task, i.e., tasks with minimal memory demand. The paper is overall well written and easy to read. The figures are clear and the statisticical approach for the critical analysis seem sound. Below I outline major and minor questions.

#### Major:

- 1. If I understand correctly, in Experiment 1 the SR were familiar with more of the famous faces than the control groups (i.e, SR: 11/12 versus controls 9/12 = 92% vs 75%). Could this also explain why the SR were better in the actual test (Figure 1)? That is, do they score higher because they are super recognizers or do they score higher because they happened to be more familiar with the people? If the latter, it would imply that super recognizers are only perhaps super recognizers for unknown faces (or in other words, everyone is super recognizers with familiar faces)?
- 2. For Experiments 1-3, what does the RT data look like? Could the increased accuracy for super recognizers be explained by also longer RTs, or are they both more accurate and faster?
- 3. For the interactions in Experiment 3, the authors choose to focus on the emotion x disguise rather than the emotion x group (or both). A rationale for why one is analyzed, but not the other, could be useful. Since the graph splits the data by group, it seems strange to skip any analysis where group is involved.
- 4. In the introduction, it could be useful to be more specific about the novel aspects of this study. For example, perhaps a brief summary sentence(s) in the final paragraph could state the novelty (and perhaps a rationale for why it is important)? The intro does mention novel aspects throughout, but some places it is a bit unclear what exactly is novel. For example, on the second page of the introduction (first paragraph), it is referred to research that occludes eyes with sunglasses [30], and another (preprint) study testing the effects of surgical masks [37], yet the concluding sentence of the paragraph reads "To date, no studies have compared the effect of occluding the eye region and occluding the mouth region on face matching performance". In this particular example, is novelty the face matching task (rather than some other recognition task) or is it that both conditions are included in the same study. In either case, a rationale for why this is important could also help the readers.
- 5. Practiced participants: When definining the 'practiced participants', the MS states " [Practiced participants] appeared to be particularly engaged and continued to take part in various face perception studies"? What does "particularly engaged" mean? How many experiments on average had the Ps already taken part in? Did they receive any face training during these experiments (or feedback for correct/incorrect trials)? If they only participated in standard experiments without feedback or training, then is there a rational reason to believe that they have become better at face recognition as a result of these studies (I can see that they are better at grasping the mechanics of a computerized face recognition task, but whether they have actually become better in face recognition is more difficult to apprhend)?

- 6. In the last sentence of stimuli section for Experiment 1, it states that Ps "saw two trials in each concealment condition (....) for each trial type". Is it correct that there were only 4 trials per condition (2 match and 2 nonmatch), or was this per face identity (thus, 4\*12 trials). If the latter, did face images repeat for each celebrity? If the former, why so few trials? Relatedly, it would be good to explicitly state whether each identity was shown in each condition.
- 7. For the correlation data (Figure 4, 5), why were not the non-practiced participants added?
- 8. In the summary for Experiment 2, it is stated that one could expect no overlap between SR and practiced controls. However, given that experiments are noisy (especially true for short experiments), I would in fact have expected that some super recognizers would overlap with practiced controls just due to chance, yet that they will be different at the group level. AlsoThat said, if the arguement is kept, then it could be benefical to report the range of scores on the CFMT. Based on the argument, there should be no overlap between the highest practiced participant and lowest SR (unless I mistook something)?

#### Minor

9. page 4 (intro): "... task difficulty (e.g., [21] and ...." --> bracket missing?

- 10. The introduction refers to findings with memory tasks, but a lack of similar findings with face matching tasks (e.g., page 2 of introduction, 2nd paragraph). It could be useful to be more specific about the memory task, if the goal is to distinguish the face matching task from the memory task. For example, what was the memory demand of the memory task (e.g., 1 second, 10 second, 30 minutes)? Note that even in a face matching task, there is a memory demand, since once can (presumably) only fixate/attend one face at the time.
- 11. In E1, perhaps parts of the stimuli section referring to trial structure would be more suited in the subsequent procedure section? That is, the trial structure is meaningless without knowing the paradigm, which is described in the following 'procedure' section.
- 12. Experiment 1 procedure. How long were the displays on for? Timed or until response? Would be good to be more specific about the procedure, and perhaps even add a figure (e.g., as panel A for Figure 1).
- 13. Experiment 1, results: When referring to number of celebrities known, it could be good to use percentage or at least show the maximum possible (e.g., M = 75% (9/12)). Also, if this is a mean score, it seems strange that there are no decimals?

Review form: Reviewer 2

Is the manuscript scientifically sound in its present form? No

Are the interpretations and conclusions justified by the results? Yes

Is the language acceptable?

Do you have any ethical concerns with this paper?

No

# Have you any concerns about statistical analyses in this paper?

#### Recommendation?

Major revision is needed (please make suggestions in comments)

#### Comments to the Author(s)

In this manuscript the authors pursue the contemporary / obvious question of whether face occlusion impairs facial identity matching and emotion categorization, and whether such an impairment varies as a function of independently assessed face matching / recognition performance. Their results are as anticipated: less information available impairs performance (experiments 1 & 2), and more so if the information occluded is that which is diagnostic for the task (experiment 3). Below I detail main points, which I believe can be addressed. Sincerely,

Meike Ramon

#### Analyses

The authors should fully exploit the fact that their observers performed all tasks, and include a more detailed, multivariate consideration of their performance patterns across all three experiments, under consideration of patterns at the individual level as they propose. Appropriate visualisations should also be provided in order for readers to fully appreciate the depth of available information at the single-subject level.

Correlational analyses should be provided throughout (not only selectively), as well as with/without inclusion of supers to address the question of whether they are driving linear relationships.

p.19., 2nd paragraph and Table 1: extremely cumbersome to work through; provide more straightforward approach and reporting.

#### Missing methodological information

There is no information provided regarding quality assurance measures implemented to control for viewing conditions across observers, which is particularly important given data acquisition was performed online. I would like to see the required information considering restrictions implemented to prevent participation from tablets and mobile phones, as well as details regarding implemented screen calibration procedures.

No detail is provided regarding the selection of image material for experiments 1 and 2 thereby preventing reproduction.

Describe how authors "ensure[d] that the intensity and validity of expressions was consistent across concealment conditions".

#### Diagnostic approach at odds with authors own recent practice

In a recent manuscript (Dunn et al., 2020) the GFMT and CFMT+ were described as inappropriate for online assessment, motivating the creation of an(other) online test battery. Therefore, instead of framing the manuscript as a "Super-Recognizer" study the more appropriate approach is to consider only the measure which does not suffer from a ceiling effect (ie the CFMT+), and to investigate the relationship between face recognition performance and the manipulations implemented here.

#### Missing references

A number of relevant references, which should be cited are missing. These relate to the issue of familiar face processing including the aspect of partial information processing (cf. Ramon & Gobbini, 2018, Vis Cogn; Ramon, Busigny, Gosselin & Rossion, 2018, Vis Cogn), as well as those concerning the relationship across tests observed at the group vs. individual level (Stacchi et al.,

2020; Fysh et al., 2020), and notably the review of super-recogniser literature by Ramon et al., 2019 (a,b; BJP).

#### Minor points

- identification should not be used interchangeably with recognition and matching (eg first sentence in the abstract), and in light of the reported manipulations implemented here refer to the tasks with greater precision (e.g. p.4, l.31: comparisons of facial \*identity\*; given the nature of response measurement experiment 3 is actually an emotion categorisation, rather than recognition task).
- p.5, 1st paragraph: the related composite face effect should be discussed, as well as its implications that facial information in the bottom of the face may have an either facilitatory as well as inhibitory effect.
- remove p.5 2nd paragraph; eye tracking in normal controls and supers not appropriate, as information sampling does not imply usage, and the prior is not assessed here.
- range of performance scores should be provided for all samples reported (cf. p.8)
- remove reference to results reported purely in the Supplementary Material (cf. p.11, l.17f.)

### Decision letter (RSOS-201169.R0)

We hope you are keeping well at this difficult and unusual time. We continue to value your support of the journal in these challenging circumstances. If Royal Society Open Science can assist you at all, please don't hesitate to let us know at the email address below.

#### Dear Dr Noyes

The Editors assigned to your paper RSOS-201169 "The effect of face masks and sunglasses on identity and expression recognition with super-recognisers and typical observers" have now received comments from reviewers and would like you to revise the paper in accordance with the reviewer comments and any comments from the Editors. Please note this decision does not guarantee eventual acceptance.

We invite you to respond to the comments supplied below and revise your manuscript. Below the referees' and Editors' comments (where applicable) we provide additional requirements. Final acceptance of your manuscript is dependent on these requirements being met. We provide guidance below to help you prepare your revision.

We do not generally allow multiple rounds of revision so we urge you to make every effort to fully address all of the comments at this stage. If deemed necessary by the Editors, your manuscript will be sent back to one or more of the original reviewers for assessment. If the original reviewers are not available, we may invite new reviewers.

Please submit your revised manuscript and required files (see below) no later than 21 days from today's (ie 05-Nov-2020) date. Note: the ScholarOne system will 'lock' if submission of the revision is attempted 21 or more days after the deadline. If you do not think you will be able to meet this deadline please contact the editorial office immediately.

Please note article processing charges apply to papers accepted for publication in Royal Society Open Science (https://royalsocietypublishing.org/rsos/charges). Charges will also apply to papers transferred to the journal from other Royal Society Publishing journals, as well as papers submitted as part of our collaboration with the Royal Society of Chemistry

(https://royalsocietypublishing.org/rsos/chemistry). Fee waivers are available but must be requested when you submit your revision (https://royalsocietypublishing.org/rsos/waivers).

Thank you for submitting your manuscript to Royal Society Open Science and we look forward to receiving your revision. If you have any questions at all, please do not hesitate to get in touch.

Kind regards, Anita Kristiansen Editorial Coordinator

Royal Society Open Science openscience@royalsociety.org

on behalf of Dr Bruno Rossion (Associate Editor) and Essi Viding (Subject Editor) openscience@royalsociety.org

Reviewer comments to Author:

Reviewer: 1

Comments to the Author(s)

Noyes et al. examined the extent to which recognition of famous face identity (E1), unfamiliar face identity (E2), and face emotion (E3) is influenced by disguising the eyes with sunglasses or the mouth with masks. This was compared in people with typical recognition ability and so-called super recognizers (SR). Recognition performance was measured in a face matching task (E1-2) and a forced-choice task, i.e., tasks with minimal memory demand. The paper is overall well written and easy to read. The figures are clear and the statisticical approach for the critical analysis seem sound. Below I outline major and minor questions.

#### Major:

- 1. If I understand correctly, in Experiment 1 the SR were familiar with more of the famous faces than the control groups (i.e, SR: 11/12 versus controls 9/12 = 92% vs 75%). Could this also explain why the SR were better in the actual test (Figure 1)? That is, do they score higher because they are super recognizers or do they score higher because they happened to be more familiar with the people? If the latter, it would imply that super recognizers are only perhaps super recognizers for unknown faces (or in other words, everyone is super recognizers with familiar faces)?
- 2. For Experiments 1-3, what does the RT data look like? Could the increased accuracy for super recognizers be explained by also longer RTs, or are they both more accurate and faster?
- 3. For the interactions in Experiment 3, the authors choose to focus on the emotion x disguise rather than the emotion x group (or both). A rationale for why one is analyzed, but not the other, could be useful. Since the graph splits the data by group, it seems strange to skip any analysis where group is involved.
- 4. In the introduction, it could be useful to be more specific about the novel aspects of this study. For example, perhaps a brief summary sentence(s) in the final paragraph could state the novelty (and perhaps a rationale for why it is important)? The intro does mention novel aspects throughout, but some places it is a bit unclear what exactly is novel. For example, on the second page of the introduction (first paragraph), it is referred to research that occludes eyes with sunglasses [30], and another (preprint) study testing the effects of surgical masks [37], yet the

concluding sentence of the paragraph reads "To date, no studies have compared the effect of occluding the eye region and occluding the mouth region on face matching performance". In this particular example, is novelty the face matching task (rather than some other recognition task) or is it that both conditions are included in the same study. In either case, a rationale for why this is important could also help the readers.

- 5. Practiced participants: When definining the 'practiced participants', the MS states " [Practiced participants] appeared to be particularly engaged and continued to take part in various face perception studies"? What does "particularly engaged" mean? How many experiments on average had the Ps already taken part in? Did they receive any face training during these experiments (or feedback for correct/incorrect trials)? If they only participated in standard experiments without feedback or training, then is there a rational reason to believe that they have become better at face recognition as a result of these studies (I can see that they are better at grasping the mechanics of a computerized face recognition task, but whether they have actually become better in face recognition is more difficult to apprhend)?
- 6. In the last sentence of stimuli section for Experiment 1, it states that Ps "saw two trials in each concealment condition (....) for each trial type". Is it correct that there were only 4 trials per condition (2 match and 2 nonmatch), or was this per face identity (thus, 4\*12 trials). If the latter, did face images repeat for each celebrity? If the former, why so few trials? Relatedly, it would be good to explicitly state whether each identity was shown in each condition.
- 7. For the correlation data (Figure 4, 5), why were not the non-practiced participants added?
- 8. In the summary for Experiment 2, it is stated that one could expect no overlap between SR and practiced controls. However, given that experiments are noisy (especially true for short experiments), I would in fact have expected that some super recognizers would overlap with practiced controls just due to chance, yet that they will be different at the group level. AlsoThat said, if the arguement is kept, then it could be benefical to report the range of scores on the CFMT. Based on the argument, there should be no overlap between the highest practiced participant and lowest SR (unless I mistook something)?

#### Minor

9. page 4 (intro): "... task difficulty (e.g., [21] and ...." --> bracket missing?

- 10. The introduction refers to findings with memory tasks, but a lack of similar findings with face matching tasks (e.g., page 2 of introduction, 2nd paragraph). It could be useful to be more specific about the memory task, if the goal is to distinguish the face matching task from the memory task. For example, what was the memory demand of the memory task (e.g., 1 second, 10 second, 30 minutes)? Note that even in a face matching task, there is a memory demand, since once can (presumably) only fixate/attend one face at the time.
- 11. In E1, perhaps parts of the stimuli section referring to trial structure would be more suited in the subsequent procedure section? That is, the trial structure is meaningless without knowing the paradigm, which is described in the following 'procedure' section.
- 12. Experiment 1 procedure. How long were the displays on for? Timed or until response? Would be good to be more specific about the procedure, and perhaps even add a figure (e.g., as panel A for Figure 1).
- 13. Experiment 1, results: When referring to number of celebrities known, it could be good to use percentage or at least show the maximum possible (e.g., M = 75% (9/12)). Also, if this is a mean score, it seems strange that there are no decimals?

#### Reviewer: 2

#### Comments to the Author(s)

In this manuscript the authors pursue the contemporary / obvious question of whether face occlusion impairs facial identity matching and emotion categorization, and whether such an impairment varies as a function of independently assessed face matching / recognition performance. Their results are as anticipated: less information available impairs performance (experiments 1 & 2), and more so if the information occluded is that which is diagnostic for the task (experiment 3). Below I detail main points, which I believe can be addressed. Sincerely,

Meike Ramon

#### Analyses

The authors should fully exploit the fact that their observers performed all tasks, and include a more detailed, multivariate consideration of their performance patterns across all three experiments, under consideration of patterns at the individual level as they propose. Appropriate visualisations should also be provided in order for readers to fully appreciate the depth of available information at the single-subject level.

Correlational analyses should be provided throughout (not only selectively), as well as with/without inclusion of supers to address the question of whether they are driving linear relationships.

p.19., 2nd paragraph and Table 1: extremely cumbersome to work through; provide more straightforward approach and reporting.

#### Missing methodological information

There is no information provided regarding quality assurance measures implemented to control for viewing conditions across observers, which is particularly important given data acquisition was performed online. I would like to see the required information considering restrictions implemented to prevent participation from tablets and mobile phones, as well as details regarding implemented screen calibration procedures.

No detail is provided regarding the selection of image material for experiments 1 and 2 thereby preventing reproduction.

Describe how authors "ensure[d] that the intensity and validity of expressions was consistent across concealment conditions".

#### Diagnostic approach at odds with authors own recent practice

In a recent manuscript (Dunn et al., 2020) the GFMT and CFMT+ were described as inappropriate for online assessment, motivating the creation of an(other) online test battery. Therefore, instead of framing the manuscript as a "Super-Recognizer" study the more appropriate approach is to consider only the measure which does not suffer from a ceiling effect (ie the CFMT+), and to investigate the relationship between face recognition performance and the manipulations implemented here.

#### Missing references

A number of relevant references, which should be cited are missing. These relate to the issue of familiar face processing including the aspect of partial information processing (cf. Ramon & Gobbini, 2018, Vis Cogn; Ramon, Busigny, Gosselin & Rossion, 2018, Vis Cogn), as well as those concerning the relationship across tests observed at the group vs. individual level (Stacchi et al., 2020; Fysh et al., 2020), and notably the review of super-recogniser literature by Ramon et al., 2019 (a,b; BJP).

#### Minor points

- identification should not be used interchangeably with recognition and matching (eg first sentence in the abstract), and in light of the reported manipulations implemented here refer to the tasks with greater precision (e.g. p.4, l.31: comparisons of facial \*identity\*; given the nature of response measurement experiment 3 is actually an emotion categorisation, rather than recognition task).
- p.5, 1st paragraph: the related composite face effect should be discussed, as well as its implications that facial information in the bottom of the face may have an either facilitatory as well as inhibitory effect.
- remove p.5 2nd paragraph; eye tracking in normal controls and supers not appropriate, as information sampling does not imply usage, and the prior is not assessed here.
- range of performance scores should be provided for all samples reported (cf. p.8)
- remove reference to results reported purely in the Supplementary Material (cf. p.11, l.17f.)

#### ===PREPARING YOUR MANUSCRIPT===

Your revised paper should include the changes requested by the referees and Editors of your manuscript. You should provide two versions of this manuscript and both versions must be provided in an editable format:

one version identifying all the changes that have been made (for instance, in coloured highlight, in bold text, or tracked changes);

a 'clean' version of the new manuscript that incorporates the changes made, but does not highlight them. This version will be used for typesetting if your manuscript is accepted. Please ensure that any equations included in the paper are editable text and not embedded images.

Please ensure that you include an acknowledgements' section before your reference list/bibliography. This should acknowledge anyone who assisted with your work, but does not qualify as an author per the guidelines at https://royalsociety.org/journals/ethics-policies/openness/.

While not essential, it will speed up the preparation of your manuscript proof if accepted if you format your references/bibliography in Vancouver style (please see https://royalsociety.org/journals/authors/author-guidelines/#formatting). You should include DOIs for as many of the references as possible.

If you have been asked to revise the written English in your submission as a condition of publication, you must do so, and you are expected to provide evidence that you have received language editing support. The journal would prefer that you use a professional language editing service and provide a certificate of editing, but a signed letter from a colleague who is a native speaker of English is acceptable. Note the journal has arranged a number of discounts for authors using professional language editing services

(https://royalsociety.org/journals/authors/benefits/language-editing/).

#### ===PREPARING YOUR REVISION IN SCHOLARONE===

To revise your manuscript, log into https://mc.manuscriptcentral.com/rsos and enter your Author Centre - this may be accessed by clicking on "Author" in the dark toolbar at the top of the page (just below the journal name). You will find your manuscript listed under "Manuscripts with Decisions". Under "Actions", click on "Create a Revision".

Attach your point-by-point response to referees and Editors at Step 1 'View and respond to decision letter'. This document should be uploaded in an editable file type (.doc or .docx are preferred). This is essential.

Please ensure that you include a summary of your paper at Step 2 'Type, Title, & Abstract'. This should be no more than 100 words to explain to a non-scientific audience the key findings of your research. This will be included in a weekly highlights email circulated by the Royal Society press office to national UK, international, and scientific news outlets to promote your work.

At Step 3 'File upload' you should include the following files:

- -- Your revised manuscript in editable file format (.doc, .docx, or .tex preferred). You should upload two versions:
- 1) One version identifying all the changes that have been made (for instance, in coloured highlight, in bold text, or tracked changes);
- 2) A 'clean' version of the new manuscript that incorporates the changes made, but does not highlight them.
- -- An individual file of each figure (EPS or print-quality PDF preferred [either format should be produced directly from original creation package], or original software format).
- -- An editable file of each table (.doc, .docx, .xls, .xlsx, or .csv).
- -- An editable file of all figure and table captions.

Note: you may upload the figure, table, and caption files in a single Zip folder.

- -- Any electronic supplementary material (ESM).
- -- If you are requesting a discretionary waiver for the article processing charge, the waiver form must be included at this step.
- -- If you are providing image files for potential cover images, please upload these at this step, and inform the editorial office you have done so. You must hold the copyright to any image provided.
- -- A copy of your point-by-point response to referees and Editors. This will expedite the preparation of your proof.

At Step 6 'Details & comments', you should review and respond to the queries on the electronic submission form. In particular, we would ask that you do the following:

- -- Ensure that your data access statement meets the requirements at https://royalsociety.org/journals/authors/author-guidelines/#data. You should ensure that you cite the dataset in your reference list. If you have deposited data etc in the Dryad repository, please include both the 'For publication' link and 'For review' link at this stage.
- -- If you are requesting an article processing charge waiver, you must select the relevant waiver option (if requesting a discretionary waiver, the form should have been uploaded at Step 3 'File upload' above).
- -- If you have uploaded ESM files, please ensure you follow the guidance at https://royalsociety.org/journals/authors/author-guidelines/#supplementary-material to include a suitable title and informative caption. An example of appropriate titling and captioning may be found at https://figshare.com/articles/Table\_S2\_from\_Is\_there\_a\_trade-off\_between\_peak\_performance\_and\_performance\_breadth\_across\_temperatures\_for\_aerobic\_sc ope\_in\_teleost\_fishes\_/3843624.

At Step 7 'Review & submit', you must view the PDF proof of the manuscript before you will be able to submit the revision. Note: if any parts of the electronic submission form have not been completed, these will be noted by red message boxes.

# Author's Response to Decision Letter for (RSOS-201169.R0)

# RSOS-201169.R1 (Revision)

Review form: Reviewer 1

Is the manuscript scientifically sound in its present form?

Yes

Are the interpretations and conclusions justified by the results?

Yes

Is the language acceptable?

Yes

Do you have any ethical concerns with this paper?

No

Have you any concerns about statistical analyses in this paper?

No

#### Recommendation?

Major revision is needed (please make suggestions in comments)

#### Comments to the Author(s)

I thank the authors for their responses and revisions. Below are my remaining comments.

- Response to comments #1 and #2: I think these limitations should at least be explicitly mentioned in the general discussion. Essentially, the enhanced performance of super recognizers in E1 could be due to they simply being more familiar with the faces (i.e., not specific to super recognizer ability). Moreover, the enhanced performance of super recognizers in general could be due to them spending more time (e.g., being more meticulous) than the other participants on each image before making their judgement.
- Response to comment #5: Perhaps explicitly define practiced and non-practiced in the text. E.g., (if I understand the definition of each group correctly)
- o practiced: participated in previous experiments from the lab
- o non-practiced: randomly recruited Ps who may or may not have done face recognition studies in the past.

Moreover, it seems unclear whether the "practiced" participants have in fact completed more face-recognition studies than the "non-practiced" participants? If I understand correctly, the practiced have done more face recognition tests offered by this specific research group, but it seems possible that the non-practiced Ps could have completed face recognition tests from other groups? If there is no way to ascertain this, then I think this limitation should also be explicitly mentioned in the discussion.

- Response to comment #6: Given the few trials (which seems abnormal(?)), this might also be good to point out as a potential limitation in the discussion.

# Decision letter (RSOS-201169.R1)

We hope you are keeping well at this difficult and unusual time. We continue to value your support of the journal in these challenging circumstances. If Royal Society Open Science can assist you at all, please don't hesitate to let us know at the email address below.

#### Dear Dr Noyes

The Editors assigned to your paper RSOS-201169.R1 "The effect of face masks and sunglasses on identity and expression recognition with super-recognisers and typical observers" have now received comments from reviewers and would like you to revise the paper in accordance with the reviewer comments and any comments from the Editors. Please note this decision does not guarantee eventual acceptance.

We invite you to respond to the comments supplied below and revise your manuscript. Below the referees' and Editors' comments (where applicable) we provide additional requirements. Final acceptance of your manuscript is dependent on these requirements being met. We provide guidance below to help you prepare your revision.

We do not generally allow multiple rounds of revision so we urge you to make every effort to fully address all of the comments at this stage, given that the Editors have opted to issue a further revision - indeed, if the Editors and reviewer are not satisfied with your revision it is unlikely a further revision will be offered. If deemed necessary by the Editors, your manuscript will be sent back to one or more of the original reviewers for assessment. If the original reviewers are not available, we may invite new reviewers.

Please submit your revised manuscript and required files (see below) no later than 21 days from today's (ie 05-Feb-2021) date. Note: the ScholarOne system will 'lock' if submission of the revision is attempted 21 or more days after the deadline. If you do not think you will be able to meet this deadline please contact the editorial office immediately.

Please note article processing charges apply to papers accepted for publication in Royal Society Open Science (https://royalsocietypublishing.org/rsos/charges). Charges will also apply to papers transferred to the journal from other Royal Society Publishing journals, as well as papers submitted as part of our collaboration with the Royal Society of Chemistry (https://royalsocietypublishing.org/rsos/chemistry). Fee waivers are available but must be requested when you submit your revision (https://royalsocietypublishing.org/rsos/waivers).

Thank you for submitting your manuscript to Royal Society Open Science and we look forward to receiving your revision. If you have any questions at all, please do not hesitate to get in touch.

Kind regards, Royal Society Open Science Editorial Office Royal Society Open Science openscience@royalsociety.org

on behalf of Dr Bruno Rossion (Associate Editor) and Essi Viding (Subject Editor) openscience@royalsociety.org

Reviewer comments to Author:

Reviewer: 1

Comments to the Author(s)

I thank the authors for their responses and revisions. Below are my remaining comments.

- Response to comments #1 and #2: I think these limitations should at least be explicitly mentioned in the general discussion. Essentially, the enhanced performance of super recognizers in E1 could be due to they simply being more familiar with the faces (i.e., not specific to super recognizer ability). Moreover, the enhanced performance of super recognizers in general could be due to them spending more time (e.g., being more meticulous) than the other participants on each image before making their judgement.
- Response to comment #5: Perhaps explicitly define practiced and non-practiced in the text. E.g., (if I understand the definition of each group correctly)

o practiced: participated in previous experiments from the lab o non-practiced: randomly recruited Ps who may or may not have done face recognition studies in the past.

Moreover, it seems unclear whether the "practiced" participants have in fact completed more face-recognition studies than the "non-practiced" participants? If I understand correctly, the practiced have done more face recognition tests offered by this specific research group, but it seems possible that the non-practiced Ps could have completed face recognition tests from other groups? If there is no way to ascertain this, then I think this limitation should also be explicitly mentioned in the discussion.

- Response to comment #6: Given the few trials (which seems abnormal(?)), this might also be good to point out as a potential limitation in the discussion.

#### ===PREPARING YOUR MANUSCRIPT===

Your revised paper should include the changes requested by the referees and Editors of your manuscript. You should provide two versions of this manuscript and both versions must be provided in an editable format:

one version identifying all the changes that have been made (for instance, in coloured highlight, in bold text, or tracked changes);

a 'clean' version of the new manuscript that incorporates the changes made, but does not highlight them. This version will be used for typesetting if your manuscript is accepted. Please ensure that any equations included in the paper are editable text and not embedded images.

Please ensure that you include an acknowledgements' section before your reference list/bibliography. This should acknowledge anyone who assisted with your work, but does not qualify as an author per the guidelines at https://royalsociety.org/journals/ethics-policies/openness/.

While not essential, it will speed up the preparation of your manuscript proof if accepted if you format your references/bibliography in Vancouver style (please see https://royalsociety.org/journals/authors/author-guidelines/#formatting). You should include DOIs for as many of the references as possible.

If you have been asked to revise the written English in your submission as a condition of publication, you must do so, and you are expected to provide evidence that you have received language editing support. The journal would prefer that you use a professional language editing service and provide a certificate of editing, but a signed letter from a colleague who is a native speaker of English is acceptable. Note the journal has arranged a number of discounts for authors using professional language editing services

(https://royalsociety.org/journals/authors/benefits/language-editing/).

#### ===PREPARING YOUR REVISION IN SCHOLARONE===

To revise your manuscript, log into https://mc.manuscriptcentral.com/rsos and enter your Author Centre - this may be accessed by clicking on "Author" in the dark toolbar at the top of the page (just below the journal name). You will find your manuscript listed under "Manuscripts with Decisions". Under "Actions", click on "Create a Revision".

Attach your point-by-point response to referees and Editors at Step 1 'View and respond to decision letter'. This document should be uploaded in an editable file type (.doc or .docx are preferred). This is essential.

Please ensure that you include a summary of your paper at Step 2 'Type, Title, & Abstract'. This should be no more than 100 words to explain to a non-scientific audience the key findings of your research. This will be included in a weekly highlights email circulated by the Royal Society press office to national UK, international, and scientific news outlets to promote your work.

At Step 3 'File upload' you should include the following files:

- -- Your revised manuscript in editable file format (.doc, .docx, or .tex preferred). You should upload two versions:
- 1) One version identifying all the changes that have been made (for instance, in coloured highlight, in bold text, or tracked changes);
- 2) A 'clean' version of the new manuscript that incorporates the changes made, but does not highlight them.
- -- An individual file of each figure (EPS or print-quality PDF preferred [either format should be produced directly from original creation package], or original software format).
- -- An editable file of each table (.doc, .docx, .xls, .xlsx, or .csv).
- -- An editable file of all figure and table captions.

Note: you may upload the figure, table, and caption files in a single Zip folder.

- -- Any electronic supplementary material (ESM).
- -- If you are requesting a discretionary waiver for the article processing charge, the waiver form must be included at this step.
- -- If you are providing image files for potential cover images, please upload these at this step, and inform the editorial office you have done so. You must hold the copyright to any image provided.
- -- A copy of your point-by-point response to referees and Editors. This will expedite the preparation of your proof.

At Step 6 'Details & comments', you should review and respond to the queries on the electronic submission form. In particular, we would ask that you do the following:

- -- Ensure that your data access statement meets the requirements at https://royalsociety.org/journals/authors/author-guidelines/#data. You should ensure that you cite the dataset in your reference list. If you have deposited data etc in the Dryad repository, please include both the 'For publication' link and 'For review' link at this stage.
- -- If you are requesting an article processing charge waiver, you must select the relevant waiver option (if requesting a discretionary waiver, the form should have been uploaded at Step 3 'File upload' above).

-- If you have uploaded ESM files, please ensure you follow the guidance at https://royalsociety.org/journals/authors/author-guidelines/#supplementary-material to include a suitable title and informative caption. An example of appropriate titling and captioning may be found at https://figshare.com/articles/Table\_S2\_from\_Is\_there\_a\_trade-off\_between\_peak\_performance\_and\_performance\_breadth\_across\_temperatures\_for\_aerobic\_sc ope in teleost fishes /3843624.

At Step 7 'Review & submit', you must view the PDF proof of the manuscript before you will be able to submit the revision. Note: if any parts of the electronic submission form have not been completed, these will be noted by red message boxes.

# Author's Response to Decision Letter for (RSOS-201169.R1)

See Appendix B.

## Decision letter (RSOS-201169.R2)

We hope you are keeping well at this difficult and unusual time. We continue to value your support of the journal in these challenging circumstances. If Royal Society Open Science can assist you at all, please don't hesitate to let us know at the email address below.

Dear Dr Noyes,

It is a pleasure to accept your manuscript entitled "The effect of face masks and sunglasses on identity and expression recognition with super-recognisers and typical observers" in its current form for publication in Royal Society Open Science. The comments of the reviewer(s) who reviewed your manuscript are included at the foot of this letter.

Please ensure that you send to the editorial office an editable version of your accepted manuscript, and individual files for each figure and table included in your manuscript. You can send these in a zip folder if more convenient. Failure to provide these files may delay the processing of your proof. You may disregard this request if you have already provided these files to the editorial office.

You can expect to receive a proof of your article in the near future. Please contact the editorial office (openscience@royalsociety.org) and the production office (openscience\_proofs@royalsociety.org) to let us know if you are likely to be away from e-mail contact -- if you are going to be away, please nominate a co-author (if available) to manage the proofing process, and ensure they are copied into your email to the journal.

Due to rapid publication and an extremely tight schedule, if comments are not received, your paper may experience a delay in publication.

Please see the Royal Society Publishing guidance on how you may share your accepted author manuscript at https://royalsociety.org/journals/ethics-policies/media-embargo/. After publication, some additional ways to effectively promote your article can also be found here https://royalsociety.org/blog/2020/07/promoting-your-latest-paper-and-tracking-your-results/.

Thank you for your fine contribution. On behalf of the Editors of Royal Society Open Science, we look forward to your continued contributions to the Journal.

Kind regards, Anita Kristiansen Editorial Coordinator

Royal Society Open Science openscience@royalsociety.org

on behalf of Dr Bruno Rossion (Associate Editor) and Essi Viding (Subject Editor) openscience@royalsociety.org

Follow Royal Society Publishing on Twitter: @RSocPublishing Follow Royal Society Publishing on Facebook: https://www.facebook.com/RoyalSocietyPublishing.FanPage/ Read Royal Society Publishing's blog: https://royalsociety.org/blog/blogsearchpage/?category=Publishing

# **Appendix A**



RSOS-201168

Royal Society Open Science

Title: The effect of face masks and sunglasses on identity and expression recognition with super-recognisers and typical observers.

Authors: Eilidh Noyes, Josh Davis, Nikolay Petrov, Katie L.H. Gray, Kay L. Ritchie.

26/11/20

Dear Editor,

Please find uploaded the revised version of the manuscript 'The effect of face masks and sunglasses on identity and expression recognition with super-recognisers and typical observers' with co-authors Josh Davis, Nikolay Petrov, Katie L.H. Gray and Kay L. Ritchie. We hope that you will consider this revised paper for publication in the journal Royal Society Open Science.

We would like to thank you and the reviewers for the comments on the manuscript, which we have addressed in turn.

Please do not hesitate to contact me if I may be of any assistance in processing the manuscript.

Yours Sincerely,

Eilidh Noyes

Eilidh Noyes, Ph.D.

Dept. of Psychology

University of Huddersfield

Huddersfield, UK

HD13DH

Email: e.noyes@hud.ac.uk

Tel: 01484472770

#### RSOS-201168

Royal Society Open Science

Title: The effect of face masks and sunglasses on identity and expression recognition with super-recognisers and typical observers

Authors: Eilidh Noyes, Josh Davis, Nikolay Petrov, Katie L.H. Gray, Kay L. Ritchie.

# Reviewer comments to Author: Reviewer #1:

Comments to the Author(s)

Noyes et al. examined the extent to which recognition of famous face identity (E1), unfamiliar face identity (E2), and face emotion (E3) is influenced by disguising the eyes with sunglasses or the mouth with masks. This was compared in people with typical recognition ability and so-called super recognizers (SR). Recognition performance was measured in a face matching task (E1-2) and a forced-choice task, i.e., tasks with minimal memory demand. The paper is overall well written and easy to read. The figures are clear and the statistical approach for the critical analysis seem sound. Below I outline major and minor questions.

**Comment 1.** If I understand correctly, in Experiment 1 the SR were familiar with more of the famous faces than the control groups (i.e, SR: 11/12 versus controls 9/12 = 92% vs 75%). Could this also explain why the SR were better in the actual test (Figure 1)? That is, do they score higher because they are super recognizers or do they score higher because they happened to be more familiar with the people? If the latter, it would imply that super recognizers are only perhaps super recognizers for unknown faces (or in other words, everyone is super recognizers with familiar faces)?

- **Response.** We have added text to address this comment on page 11 "This familiarity advantage may explain the difference in performance of SRs and controls".

**Comment 2.** For Experiments 1-3, what does the RT data look like? Could the increased accuracy for super recognizers be explained by also longer RTs, or are they both more accurate and faster?

- **Response.** We did not record reaction time data for Experiments 1-3. We have suggested on page 23 that although group RT differences between super-recognisers and control participants have not been found in memory tasks (Belanova et al., 2018), this could be an interesting avenue for future research.

**Comment 3**. For the interactions in Experiment 3, the authors choose to focus on the emotion x disguise rather than the emotion x group (or both). A rationale for why one is analyzed, but not the other, could be useful. Since the graph splits the data by group, it seems strange to skip any analysis where group is involved.

- **Response.** We now report the breakdown for both the emotion x disguise and the emotion x group interactions.

Comment 4. In the introduction, it could be useful to be more specific about the novel

aspects of this study. For example, perhaps a brief summary sentence(s) in the final paragraph could state the novelty (and perhaps a rationale for why it is important)? The intro does mention novel aspects throughout, but some places it is a bit unclear what exactly is novel. For example, on the second page of the introduction (first paragraph), it is referred to research that occludes eyes with sunglasses [30], and another (preprint) study testing the effects of surgical masks [37], yet the concluding sentence of the paragraph reads "To date, no studies have compared the effect of occluding the eye region and occluding the mouth region on face matching performance". In this particular example, is novelty the face matching task (rather than some other recognition task) or is it that both conditions are included in the same study. In either case, a rationale for why this is important could also help the readers.

- Response. We have added a statement to emphasise the novelty of the current work to the end of the introduction. "In sum, here we provide the first study to directly compare face matching and emotion categorisation performance for super-recognisers and typical observers for faces in no concealment, sunglasses and masks. Our study is novel in its use of real mask images (rather than computer generated facial occlusions) in the matching task. Our matching task mask stimuli are representative of the types of masks worn during COVID-19." We have also clarified the text on page 4: "To date, no studies have directly compared the effect of each of these conditions (occluding the eye region compared against occluding the mouth region) on face matching performance".

Comment 5. Practiced participants: When defining the 'practiced participants', the MS states "[Practiced participants] appeared to be particularly engaged and continued to take part in various face perception studies"? What does "particularly engaged" mean? How many experiments on average had the Ps already taken part in? Did they receive any face training during these experiments (or feedback for correct/incorrect trials)? If they only participated in standard experiments without feedback or training, then is there a rational reason to believe that they have become better at face recognition as a result of these studies (I can see that they are better at grasping the mechanics of a computerized face recognition task, but whether they have actually become better in face recognition is more difficult to apprehend)?

- Response. We have clarified our definition of our practiced participant group and removed the term "highly engaged" as we agree that this term is ambiguous. We describe these participants as 'practiced' as they have completed the CFMT+ and GFMT and at least one other face recognition test previously, and as such understand the demands of this type of research. These participants will also have been randomly invited to up to six face memory or matching projects per annum (we have also previously conducted three projects of face emotion recognition - again invitations will have been random), although due to privacy reasons we are not able to cross-reference how many they actually complete. In virtually all projects run through this database, scores are given at the end of the experiment, so participants will likely have awareness of their ability. In contrast, we describe those recruited through Prolific as 'unpracticed', as although it is likely they have completed a range of other research projects before, it is far less likely they will have previously completed face processing tasks.

**Comment 6.** In the last sentence of stimuli section for Experiment 1, it states that Ps "saw two trials in each concealment condition (....) for each trial type". Is it correct that there were only 4 trials per condition (2 match and 2 nonmatch), or was this per face identity (thus, 4\*12 trials). If the latter, did face images repeat for each celebrity? If the former, why so few trials? Relatedly, it would be good to explicitly state whether each identity was shown in each condition.

- **Response.** We have clarified this point in the manuscript on page 9. That is correct, there were only four trials in each condition for the familiar face matching task (Experiment 1). We did not repeat identities, so each of the 12 identities was only seen once by each participant (as in Experiment 2 with the unfamiliar faces). We made the decision to have a small number of trials so as not to repeat identities. The small number of identities used reflects the difficulty both in finding images of celebrities wearing face masks, and also of finding a number of celebrities who would be familiar to the majority of our participants.

**Comment 7.** For the correlation data (Figure 4, 5), why were not the non-practiced participants added?

Response. We do not have GFMT/CFMT+ scores for non-practiced participants.
These participants only completed our experimental tasks. The GFMT and CFMT+
scores for the SRs and practiced controls are taken from previous tests of these
participants.

**Comment 8.** In the summary for Experiment 2, it is stated that one could expect no overlap between SR and practiced controls. However, given that experiments are noisy (especially true for short experiments), I would in fact have expected that some super recognizers would overlap with practiced controls just due to chance, yet that they will be different at the group level. Also that said, if the argument is kept, then it could be beneficial to report the range of scores on the CFMT. Based on the argument, there should be no overlap between the highest practiced participant and lowest SR (unless I mistook something)?

- **Response.** We have removed the comment that the reviewer has referred to as we agree it was misleading and have replaced it with "This result demonstrates noise in experimental testing [56] and speaks to the theoretical issues associated with the definition and selection of SRs [45]". We report the range of scores on the CFMT and GFMT on page 14.

#### **Minor Comments**

Comment 9. page 4 (intro): "... task difficulty (e.g., [21] and ...." --> bracket missing?

**- Response.** We have corrected this typing error.

Comment 10. The introduction refers to findings with memory tasks, but a lack of similar

findings with face matching tasks (e.g., page 2 of introduction, 2nd paragraph). It could be useful to be more specific about the memory task, if the goal is to distinguish the face matching task from the memory task. For example, what was the memory demand of the memory task (e.g., 1 second, 10 second, 30 minutes)? Note that even in a face matching task, there is a memory demand, since once can (presumably) only fixate/attend one face at the time.

- **Response.** We have added the time requirements as well as further detail of the memory tasks to the text on page 5.

**Comment 11.** In E1, perhaps parts of the stimuli section referring to trial structure would be more suited in the subsequent procedure section? That is, the trial structure is meaningless without knowing the paradigm, which is described in the following 'procedure' section.

- **Response.** We have moved the information regarding the assignment of identities to conditions etc. to the procedure section as suggested.

**Comment 12.** Experiment 1 procedure. How long were the displays on for? Timed or until response? Would be good to be more specific about the procedure, and perhaps even add a figure (e.g., as panel A for Figure 1).

- **Response.** We have added text to clarify stimuli presentation duration (until response) for Experiment 1.

Comment 13. Experiment 1, results: When referring to number of celebrities known, it could be good to use percentage or at least show the maximum possible (e.g., M = 75% (9/12)). Also, if this is a mean score, it seems strange that there are no decimals?

- **Response.** We have added the percentage with decimal places, and maximum.

#### Reviewer #2.

Comments to the Author(s)

In this manuscript the authors pursue the contemporary / obvious question of whether face occlusion impairs facial identity matching and emotion categorization, and whether such an impairment varies as a function of independently assessed face matching / recognition performance. Their results are as anticipated: less information available impairs performance (experiments 1 & 2), and more so if the information occluded is that which is diagnostic for the task (experiment 3). Below I detail main points, which I believe can be addressed. Sincerely,

Meike Ramon

**Comment 1.** Analyses. The authors should fully exploit the fact that their observers performed all tasks, and include a more detailed, multivariate consideration of their performance patterns across all three experiments, under consideration of patterns at the individual level as they propose. Appropriate visualisations should also be provided in order

for readers to fully appreciate the depth of available information at the single-subject level. Correlational analyses should be provided throughout (not only selectively), as well as with/without inclusion of supers to address the question of whether they are driving linear relationships.

- Response. We have added a new section to the results 'Associations across tasks', which describes how performance on each task is related to performance on the others, with appropriate visualisations (page 21). We find little evidence that the relationships are driven by the inclusion of the SRs. We have included this additional analysis, both with the inclusion and exclusion of SRs, at the task level and the condition level in the supplementary analysis.

**Comment 2.** p.19., 2nd paragraph and Table 1: extremely cumbersome to work through; provide more straightforward approach and reporting.

**Response.** We have restructured the results for Experiment 3 to make them easier to follow, both in the text and the table.

#### **Comment 3.** Missing methodological information

There is no information provided regarding quality assurance measures implemented to control for viewing conditions across observers, which is particularly important given data acquisition was performed online. I would like to see the required information considering restrictions implemented to prevent participation from tablets and mobile phones, as well as details regarding implemented screen calibration procedures.

- Response. We did not collect information about participants' devices. Our large sample somewhat protects us from the noise associated with the relative lack of control over viewing conditions induced by online data collection. Indeed, online tests of cognitive processing can yield high-quality data, indistinguishable from that collected in the lab (e.g. see Germine et al. 2012). Given that our main question of interest is focussed on the repeated-measures variable of condition, differences in viewing conditions cannot explain the effects found.

**Comment 4.** No detail is provided regarding the selection of image material for experiments 1 and 2 thereby preventing reproduction.

- **Response**. We have added additional details of image selection, such as mainly front-facing images, and having no constraints on expression etc. for Experiments 1 and 2. These are in line with previous studies, now mentioned in the stimuli sections.

**Comment 5.** Describe how authors "ensure[d] that the intensity and validity of expressions was consistent across concealment conditions".

Response. We have made this clearer in the manuscript. The underlying facial
expression was identical in the different conditions, given that we superimposed the
masks onto the exact same images.

Comment 6. Diagnostic approach at odds with authors own recent practice In a recent manuscript (Dunn et al., 2020) the GFMT and CFMT+ were described as inappropriate for online assessment, motivating the creation of an(other) online test battery. Therefore, instead of framing the manuscript as a "Super-Recognizer" study the more appropriate approach is to consider only the measure which does not suffer from a ceiling effect (ie the CFMT+), and to investigate the relationship between face recognition performance and the manipulations implemented here.

- Response. We have added text to clarify our reasons for using the CFMT+ and GFMT criteria for defining super-recognisers and controls in our experiment on pages 7 and 8. We followed the same practice as described in three published peer-reviewed papers using similar online recruitment methods (Correll, Ma, & Davis, 2020; Davis, Bretfelean, Belanova, & Thompson, 2020; Satchell et al., 2019). To the best of our knowledge, the CFMT+ has been employed in all published research articles to attribute super-recognition ability (albeit most published research has used lower thresholds than applied here), and following best practice, we use the GFMT as a second verifying test of face recognition ability. We acknowledge that the GFMT suffers from ceiling effects, but see it is a strength of our study that we include this second verification as research has shown that some super-recognisers (as assessed by the CFMT+ alone) score surprisingly poorly on other tests of face recognition ability, clearly questioning their status as super-recognisers.

We agree that the specific wording the reviewer refers to in the Dunn et al. pre-print concerning the use of the CFMT+, the GFMT, and the new test, the UNSW is confusing. The final paper has now been published and the edited confusing statement refers to the problem that because many researchers have employed the CFMT+ and the GFMT online, there is a risk of participants taking the tests more than once (and practice effects). However, we can assure the reviewer that those participants recruited to the current research had not previously taken the CFMT+ and GFMT before, and we have included statements to this effect in the manuscript.

#### **Comment 7.** Missing references

A number of relevant references, which should be cited are missing. These relate to the issue of familiar face processing including the aspect of partial information processing (cf. Ramon & Gobbini, 2018, Vis Cogn; Ramon, Busigny, Gosselin & Rossion, 2018, Vis Cogn), as well as those concerning the relationship across tests observed at the group vs. individual level (Stacchi et al., 2020; Fysh et al., 2020), and notably the review of super-recogniser literature by Ramon et al., 2019 (a,b; BJP).

- **Response.** We thank the reviewer for highlighting some additional literature and have added the following references to our revised documents (Fysh et al. 2020; Ramon et al., 2019).

#### Minor points

**1.** identification should not be used interchangeably with recognition and matching (eg first sentence in the abstract), and – in light of the reported manipulations implemented here –

refer to the tasks with greater precision (e.g. p.4, l.31: comparisons of facial \*identity\*; given the nature of response measurement experiment 3 is actually an emotion categorisation, rather than recognition task).

- **Response.** We have revised the manuscript accordingly.
- **2.** p.5, 1st paragraph: the related composite face effect should be discussed, as well as its implications that facial information in the bottom of the face may have an either facilitatory as well as inhibitory effect.
  - **Response.** We have included an additional description of the composite face illusion and its implications on page 3.
- **3.** remove p.5 2nd paragraph; eye tracking in normal controls and supers not appropriate, as information sampling does not imply usage, and the prior is not assessed here.
  - **Response.** We have edited this paragraph to make it clear that we are referring to suggested mechanisms that are posited by the literature in this field.
- **4.** range of performance scores should be provided for all samples reported (cf. p.8)
  - **Response.** We have made this edit to the manuscript.
- **5.** remove reference to results reported purely in the Supplementary Material (cf. p.11, l.17f.)
  - **Response.** We hesitate to remove the reference to the results in the supplementary material because we believe that this reference is important to direct readers to the supplementary material.

#### Additional edits

In the first version of the manuscript we used 'super-recognisers' and its abbreviated form SRs interchangeably. For clarity, we now refer to super-recognisers throughout.

We added reference to how our findings fit with theory on the value of the eye region in face perception to our discussion section.

# **Appendix B**



RSOS-201168

Royal Society Open Science

Title: The effect of face masks and sunglasses on identity and expression recognition with super-recognisers and typical observers.

Authors: Eilidh Noyes, Josh Davis, Nikolay Petrov, Katie L.H. Gray, Kay L. Ritchie.

5/02/21

Dear Editor,

Please find uploaded the revised version of the manuscript 'The effect of face masks and sunglasses on identity and expression recognition with super-recognisers and typical observers' with co-authors Josh Davis, Nikolay Petrov, Katie L.H. Gray and Kay L. Ritchie. We hope that you will consider this revised paper for publication in the journal Royal Society Open Science.

We would like to thank the reviewer for their comments on the manuscript, which we have addressed in turn.

Please do not hesitate to contact me if I may be of any assistance in processing the manuscript.

Yours Sincerely,

Eilidh Noyes

Eilidh Noyes, Ph.D.

Dept. of Psychology

University of Huddersfield

Huddersfield, UK

HD1 3DH

Email: e.noyes@hud.ac.uk

Tel: 01484472770

#### RSOS-201168

Royal Society Open Science

Title: The effect of face masks and sunglasses on identity and expression recognition with super-recognisers and typical observers

Authors: Eilidh Noyes, Josh Davis, Nikolay Petrov, Katie L.H. Gray, Kay L. Ritchie.

#### **Reviewer comments to Author:**

**Reviewer: 1** 

Comments to the Author(s)

I thank the authors for their responses and revisions. Below are my remaining comments.

- Response to comments #1 and #2: I think these limitations should at least be explicitly mentioned in the general discussion. Essentially, the enhanced performance of super recognizers in E1 could be due to they simply being more familiar with the faces (i.e., not specific to super recognizer ability). Moreover, the enhanced performance of super recognizers in general could be due to them spending more time (e.g., being more meticulous) than the other participants on each image before making their judgement.
  - **Response.** We have added text to acknowledge these in the discussion. We had already included a sentence around the possibility of speed-accuracy tradeoffs (now highlighted; page 23).
- Response to comment #5: Perhaps explicitly define practiced and non-practiced in the text. E.g., (if I understand the definition of each group correctly)

o practiced: participated in previous experiments from the lab o non-practiced: randomly recruited Ps who may or may not have done face recognition studies in the past.

Moreover, it seems unclear whether the "practiced" participants have in fact completed more face-recognition studies than the "non-practiced" participants? If I understand correctly, the practiced have done more face recognition tests offered by this specific research group, but it seems possible that the non-practiced Ps could have completed face recognition tests from other groups? If there is no way to ascertain this, then I think this limitation should also be explicitly mentioned in the discussion.

- **Response.** We had already given an extensive description at the end of the intro (page 7, now highlighted). We have added an additional definition in the methods section (page 8), and have mentioned this as a limitation to our understanding of the effect of practice in the discussion (page 23).
- Response to comment #6: Given the few trials (which seems abnormal(?)), this might also be good to point out as a potential limitation in the discussion.
  - **Response**. We have now explicitly mentioned this in the discussion (page 23). We would like to note that adequate power can be achieved using few trials over many participants, or many trials over few participants (Baker et al., 2020). Given the effects appear robust, we don't think the relatively low number of trials is concerning.